Author's response to reviews

Title: Oxpentifylline versus placebo in the treatment of erythropoietin-resistant anaemia: a randomized controlled trial

Authors:

David W Johnson (david_johnson@health.qld.gov.au)
Carmel M Hawley (carmel_hawley@health.qld.gov.au)
Brenda Rosser (Brenda.Rosser@qimr.edu.au)
Elaine Beller (e.beller@uq.edu.au)
Charles Thompson (c.thompson3@uq.edu.au)
Robert G Fassett (Rob.Fassett@dhhs.tas.gov.au)
Paolo Ferrari (paolo.ferrari@health.wa.gov.au)
Stephen P McDonald (stephenm@anzdata.org.au)
Eugenie Pedagogos (eugenie.pedagogos@mh.org.au)
Alan Cass (acass@george.org.au)

Version: 5 Date: 29 July 2008

Author's response to reviews:

I have made the following revisions:

1. Page 4 Paragraph 1 Line 4: Typo "reated" changed to "treated."
2. Page 4, Paragraph 2 Line 3: Substituted "IL-6" for "IL6" and "IL-1" for "IL1".